Author's response to reviews

Title: Self-reported cold sensitivity in normal subjects and in patients with traumatic hand injuries or hand-arm vibration syndrome

Authors:

Ingela K Carlsson (Ingela.k.carlsson@hotmail.com)
Birgitta Rosén (birgitta.rosen@med.lu.se)
Lars B Dahlin (lars.dahlin@med.lu.se)

Version: 9 Date: 28 April 2010

Author's response to reviews:

Dear Editorial Production Team,
Everything looks OK with the following exceptions:
Table 1, the last sentence should be on the same line:
....Stockholm Workshop scale, VWF)
Table 2 and 3: If you prefer the all the text below the the bottom line of the tables could be "normal" not bold and in arial.

If it is possible for you to make these changes I don’t need to see the paper anymore.

Yours Sincerely

Ingela Carlsson